Author’s response to reviews

Title: Propensity score matching analysis of a phase II study on simultaneous modulated accelerated radiation therapy using helical tomotherapy technique in patients with nasopharyngeal carcinoma

Authors:
Lei Du (coolbile_1999@163.com)
Xinxin Zhang (xinxinzhang66@hotmail.com)
Linchun Feng (301flc@163.com)
Baolin Qu (qubl6212@sina.com)
Jing Chen (chenjing19861206@163.com)
Jun Yang (yang74008@163.com)
Haixia Liu (crystal0214@163.com)
Shouping Xu (xshp228@163.com)
Chuanbin Xie (xiechuanbin2003@163.com)
Lin Ma (malinpharm@sina.com)

Version: 3 Date: 27 Jul 2017

Author’s response to reviews:

Responses to editor comments

1. Thank you for providing the "Availability of data and materials" section. However, please provide clarification as to how the data can be accessed from http://www.medresman.org/pub/en/proj/listbyproj.aspx?proj=627.

Response: We are very sorry that we have ignored to open the datasets to the public. Now the general datasets can be consulted from [http://www.medresman.org/pub/en/proj/listbyproj.aspx?proj=627]. If necessary, the detailed data are available from the corresponding author on reasonable request.